Author's response to reviews

Title: Aqueous extract of Terminalia arjuna prevents carbon tetrachloride induced hepatic and renal disorders

Authors:

Prasenjit Manna (prasenjitmanna2005@yahoo.co.in)
Mahua Sinha (sinha.mahua@rediffmail.com)
Parames C Sil (parames@bosemain.boseinst.ac.in)

Version: 7  Date: 27 September 2006

Author's response to reviews:

September 27, 2006

Professor Emma Parkin
Associate Editor
BMC-series journals

Dear Prof. Parkin,

Thank you very much for accepting our manuscript (2nd version) [MS: 1840380899105995] for publication in "BMC-Complementary and Alternative Medicine".

Following your suggestion, all the corrections have been made appropriately as mentioned below:

1. Sentence case has been used for the country name in the affiliation address.
2. Abstract- "renal damages" has been changed to "renal damage" and "Present study" has been changed to "The present study".
3. Authors' contributions- "All authors read and approved the final manuscript" has been included.
4. Acknowledgement- "Source(s) of funding" has been included.
5. Necessary changes for the figure titles and figures have been made.

Kindly acknowledge the receipt of the revised (3rd version) manuscript. I look forward to see the published manuscript in the journal.

Best regards

Dr. Parames C. Sil
Reader
Department of Chemistry
Bose Institute
93/1, A.P.C. Road
Calcutta-700009
West Bengal, INDIA
Phone : 9133-23506619/2402/2403; ext 412(lab), 413(office)FAX : 9133-2350-6790
E-mail: parames@bosemain.boseinst.ac.in / parames_95@yahoo.co.in